Author's response to reviews

Title: The Primary Care Practitioner and the diagnosis of occupational diseases

Authors:

Luca Cegolon (mrcegolon@hotmail.com)  
John H Lange (jhlange1@hotmail.com)  
Giuseppe Mastrangelo (giuseppe.mastrangelo@unipd.it)

Version: 8  Date: 8 July 2010

Author's response to reviews:

Dear BMC Production Office,

please find this new version with the revisions requested.

Particularly:

- The initials in the title have been changed to lower case character, except for Primary Care Practitioner;
- The highlightings have been removed from the body text;
- the word count has been removed from the title page;
- the city has been included in the affiliation details;
- the emails of the other 2 authors have been included in the title page;
- The body text has been structured as requested: Background; Discussion; Summary.
- The paragraph "A historical judgement by the Venice..." has been placed within the body text, just after the heading "Background";
- The sentence: "all authors read and approved the final manuscript" has been placed in the acknowledgement section at the end of the body text;
- An abbreviation section has been introduced right after the end of the manuscript and before the "competing interests".
- The table has been named "table 1" and has been referred so in the body text;
- The key word have been removed from underneath the abstract;

Thank you for your suggestion and I look forward to hearing from you soon.

Best Wishes.